UNIVERSITY BIRMINGHAM University of Birmingham Research at Birmingham

Cocaine-induced granulomatosis with polyangiitisan under-recognized condition

Gill, Charn; Sturman, Joseph; Ozbek, Leyla; Henderson, Scott R; Burns, Aine; Hamour, Sally; Pepper, Ruth J; McClelland, Lisha; Chanouzas, Dimitrios; Gane, Simon; Salama, Alan D; Harper, Lorraine

DOI: 10.1093/rap/rkad027

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Gill, C, Sturman, J, Ozbek, L, Henderson, SR, Burns, A, Hamour, S, Pepper, RJ, McClelland, L, Chanouzas, D, Gane, S, Salama, AD & Harper, L 2023, 'Cocaine-induced granulomatosis with polyangiitis-an under-recognized condition', *Rheumatology Advances in Practice*, vol. 7, no. 1, rkad027. https://doi.org/10.1093/rap/rkad027

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?) •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.



Clinical science

Cocaine-induced granulomatosis with polyangiitis—an under-recognized condition

Charn Gill¹, Joseph Sturman², Leyla Ozbek³, Scott R. Henderson⁴, Aine Burns², Sally Hamour², Ruth J. Pepper⁴, Lisha McClelland¹, Dimitrios Chanouzas (D)^{1,5}, Simon Gane³, Alan D. Salama⁴, Lorraine Harper (D ^{1,5,*}

¹Department of ENT Surgery, University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

²Renal Unit, University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK

³Department of ENT surgery, Royal National ENT Hospital, University College London Hospital, London, UK

⁴UCL Department of Renal Medicine, , Royal Free Hospital, London, UK

⁵Institute Applied Health Research, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

*Correspondence to: Lorraine Harper, Institute Applied Health Research, College of Medical and Dental Sciences, University of Birmingham, Birmingham B15 2TT, UK. E-mail: I.harper@bham.ac.uk

Abstract

Objectives: Cocaine and cocaine mixed with levamisole are increasingly used in the UK and result in significant direct nasal damage in addition to promoting vasculitis. Our aims were as follows: (1) to identify the main symptoms and presentation of cocaine-induced vasculitis; (2) to provide evidence regarding the best practice for the investigation and diagnosis of cocaine-induced vasculitis; and (3) to analyse the clinical outcomes of patients in order to understand the optimal management for the condition.

Methods: We performed a retrospective case series analysis of patients presenting with cocaine-induced midline destructive lesions or vasculitis compatible with granulomatosis with polyangiitis (GPA) from two large tertiary vasculitis clinics between 2016 and 2021.

Results: Forty-two patients (29 Birmingham, 13 London) with cocaine-induced midline lesions or systemic disease were identified. The median age was 41 years (range 23-66 years). Current cocaine use was common, and 20 of 23 samples provided were positive when routine urine toxicology was performed; 9 patients who denied ever using cocaine were identified as using cocaine based on urine toxicology analysis, and 11 who stated they were ex-users still tested positive. There was a high incidence of septal perforation (75%) and oronasal fistula (15%). Systemic manifestations were less common (27%), and only one patient had acute kidney injury. Fifty-six per cent of our patients were PR3-ANCA positive, with none testing positive for MPO-ANCA. Symptom remission required cocaine discontinuation even when immunosuppression was administered.

Conclusion: Patients with destructive nasal lesions, especially young patients, should have urine toxicology performed for cocaine before diagnosing GPA and considering immunosuppressive therapy. The ANCA pattern is not specific for cocaine-induced midline destructive lesions. Treatment should be focused on cocaine cessation and conservative management in the first instance in the absence of organ-threatening disease.

Lay Summary

What does this mean for patients?

Cocaine causes a form of inflammation of the blood vessels that primarily leads to a skin rash and destruction of the nasal structures. This can mimic an illness called granulomatosis with polyangiitis (GPA), in which inflammation of the blood vessels occurs for an unknown reason. There is little evidence about the best way to detect and treat cocaine-induced disease and whether it should be treated with strong immunosuppression like GPA. In our study, we reviewed data from 42 patients with cocaine-induced disease over a 5-year period. Our results show that 32% denied cocaine use but were positive on urine testing. The antibody ANCA is commonly found in patients with cocaine-induced disease and GPA; the pattern of ANCA does not differentiate between the two conditions. Patients found resolution of their symptoms only once they stopped using cocaine, and treatment with strong immunosuppression alone did not cause resolution of the symptoms. This study suggests that all patients presenting with skin rashes and nasal symptoms similar to GPA should have urine testing. Furthermore, the most important intervention for a good clinical outcome is cessation of cocaine use. This is more important than the use of immunosuppression, which could potentially harm patients using cocaine.

Keywords: cocaine, drug-induced vasculitis, granulomatosis with polyangiitis, ANCA, treatment

Key messages

- · Cocaine toxicology should be performed especially in young patients with a diagnosis of limited granulomatosis with polyangiitis.
- ANCA is common, but the subtype is not specific for cocaine-induced nasal disease.
- Treatment should focus on cocaine abstinence rather than immunosuppression.

Received: 3 January 2023. Accepted: 14 February 2023

© The Author(s) 2023. Published by Oxford University Press on behalf of the British Society for Rheumatology.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

Introduction

Cocaine is the second most commonly abused drug in the UK [1], with 2.6% of the population aged between 16 and 59 years old using it. It is known to cause significant pathology, including cocaine-induced midline destructive lesions [2] and various vasculitic lesions [3]. Over the last 30 years, there is a growing body of evidence showing that cocaine triggers the production of ANCA, leading to a clinical presentation which can mimic idiopathic granulomatosis with polyangiitis (GPA) [2, 4–6]. It remains unclear whether cocaine itself is the main instigator of the observed pathological process or whether the anti-helminthic drug levamisole, a compound commonly mixed with cocaine, is the true culprit [3]. In the USA, 60% of cocaine is adulterated with levamisole, and this figure is believed to be as high as 80% in Germany [7]. However, both cocaine and levamisole have been shown independently to cause a vasculitis, and both might be implicated in disease pathogenesis [3, 8–10].

The literature consists of case reports and case series with small populations of patients. The most common presentations of cocaine/levamisole-induced vasculitis are with midline destructive lesions and cutaneous vasculitis, often presenting as a retiform purpuric rash predominantly affecting the face and lower limbs [3, 11, 12]. Pulmonary and renal involvement has been reported but is rarer than the cutaneous and nasal manifestations [7, 13–15]. Although the data are limited, palatal perforation is reported as being seen only in patients who abuse cocaine and not observed in those with idiopathic GPA [11].

The reported ANCA patterns seen with cocaine-induced disease are variable, although dual PR3- and MPO-ANCA positivity has been suggested to be pathognomonic of cocaine-induced disease [16]. Anti-human neutrophil elastase antibodies have also been postulated to be strongly associated with cocaine-induced vasculitis, but routine testing for anti-human neutrophil elastase antibodies is not widely available in the UK [9].

Consequently, there remains ongoing debate about the optimal evidence-based diagnosis and management of this patient population. In this retrospective study, we report outcomes from our cohort of 42 patients to contribute additional data in order to enhance our understanding of this condition.

Methods

A retrospective review was carried out of patients presenting between 2016 and 2021 to the joint vasculitis clinic at Queen Elizabeth Hospital, Birmingham and at the Royal Free Hospital, London who disclosed cocaine use or had positive urine toxicology for cocaine with symptoms suggestive of ANCA-associated vasculitis. Data were collected retrospectively through electronic medical records. The data collected included clinical manifestations, ANCA serology by both ELISA and IIF, urine toxicology, nasal tissue histology and further management. Urine toxicology is routinely requested from patients at the Birmingham clinic on all patients referred. Urine toxicology is performed on an *ad hoc* basis dependent on levels of suspicion of cocaine use in the London clinic.

Analysis was done as part of service evaluation, registered within UHB NHS Foundation Trust under CARMS-17880.

Application of the NHS HRA research decision aid toolkit confirmed that research ethics committee approval was not required.

Results

A total of 42 patients were identified: 23 males, 18 females and 1 patient who did not identify with either gender; 13 patients from the London and 29 from the Birmingham clinics. The median age was 41 years (range 23–66 years old). The baseline demographics are shown in Table 1.

All patients were asked specifically about illicit drug use; 11 patients confirmed current cocaine use, 22 admitted to cocaine usage in the past, and 9 patients denied all cocaine use (all Birmingham). Urine toxicology is requested routinely from patients at the Birmingham clinic; 25 patients had urine toxicology performed, of whom 22 (88%) tested positive for cocaine, 9 patients denied ever using and 11 reported as exusers. Of those who tested positive, 12 individuals also tested positive for other illicit drugs or drugs of abuse, including five cannabis, four codeine, three benzodiazepines, two morphine and one levamisole. Five patients had two or more illicit substances detected in their urine in addition to cocaine. The three patients who tested negative for cocaine all confirmed that they were previous users. Of the four patients who did not provide a urine sample for toxicology, three patients confirmed previous usage but denied current use, and one stated they were currently using cocaine. Urine toxicology was performed on an ad hoc basis in the London cohort and was undertaken in nine patients, of whom seven had a positive test and confirmed current usage, while two patients confirmed previous usage and had negative urine toxicology. Of the patients from London who tested positive for cocaine, six patients also tested positive for other drugs, including three positive for opiates and four positive for cannabis.

All bar one patient had nasal symptoms. On clinical examination of those with nasal symptoms, 30 patients had evidence of septal perforation. On further examination, six patients had oronasal fistulas. Twelve patients (27%) had other systemic manifestations, which included skin lesions or rashes, joint pain, breathlessness, fatigue, diplopia and night sweats. None of our patients had pulmonary haemorrhage,

Table 1. Characteristics of the patient population

Characteristic	Birmingham (n = 29)	London (<i>n</i> = 13)
Age, median (range), years	38 (23-57)	49 (32–66)
Sex (M:F:N)	16:12:1	7:6:0
Ethnicity		
Asian	3	0
Caucasian	26	9
Black	0	0
Unknown	0	4
ANCA		
PR3-positive ELISA	17	7
MPO-positive ELISA	0	0
IIF only	8	1
pANCA by IIF and PR3	5	2
cANCA by IIF and PR3	8	1
Negative by IIF or ELISA	2	3
Not checked	2	0
Creatinine, median (range), µmol/l	74 (50–120)	77 (59–220)

F: female; M: male; N: not declared.

and median creatinine was 76 μ mol/l (range 50–220 μ mol/l). Two patients had a renal biopsy owing to haematuria, with only one patient having evidence of inflammation consistent with IgA vasculitis. This patient was the only patient with acute kidney injury (creatinine 220 μ mol/l) and had no ENT symptoms. He denied current cocaine use, with no urine toxicology performed to confirm this.

Forty patients (92.9%) had blood tests for ANCA to include MPO and PR3 antibodies measured by ELISA and cANCA and pANCA patterns measured by IIF; 87.5% (n=35) of individuals were positive by ELISA and/or IIF. Twenty-four patients (56%) tested positive for PR3 antibodies and none was positive for MPO by ELISA. Twenty-eight patients had positive ANCA results by IIF; of these, 1 was an atypical result, 11 patients were positive for cANCA and 12 for pANCA, there were 3 patients who were positive for both cANCA and pANCA and 1 reported as positive without a pattern. PR3-ANCA is usually associated with an IIF cANCA pattern; however, only nine patients positive for PR3 via ELISA had an associated cANCA pattern by IIF (37.5%) and seven (29%) PR3-positive patients were positive for p-ANCA. Four PR3-positive patients reported both cANCA and pANCA. Six PR3-positive patients were negative by IIF or borderline positive.

Twenty-eight patients had a nasal or sinus biopsy, none of whom showed evidence of granulomatous inflammation, although two patients had evidence of small vessel vasculitis.

Treatment between the two centres differed. All patients were informed of the risks of ongoing cocaine use and advised to stop. At Birmingham, patients were specifically advised that immunosuppressive treatment in the absence of organthreatening disease or nasal reconstruction was dependent on negative urine toxicology on at least two occasions. All patients were advised to commence nasal douching. Twelve of 28 patients (42.9%) were prescribed Co-trimoxazole antibiotics, and 13 of 28 (46.4%) patients were advised to use emollients. In London, patients were advised to stop using cocaine, but there was no requirement for negative toxicology before using immunosuppression and this was given according to physician preference.

Eight patients referred to the Birmingham centre had previously been given a diagnosis of GPA, with a further two patients having a differential of GPA at other centres. Of the eight patients previously diagnosed with GPA, all had been treated with immunosuppression with either one or more of rituximab, azathioprine (AZA), methotrexate (MTX), cyclophosphamide (CYC) or mycopheonlate mofetil (MMF); despite this treatment they had ongoing active disease.

With this management approach, patient follow-up of Birmingham patients was challenging. Twenty-two patients (50%) were lost to follow-up (19 did not attend and 3 were discharged back to the referring site). Seven of 29 (21%) patients remained under the care of the Birmingham joint vasculitis clinic, three of whom continue to have positive urine toxicology and symptoms, three patients have stopped cocaine usage and symptoms have settled, and one patient reports stopping but has persistent symptoms; no urine toxicology has been performed to confirm abstinence.

Of the 13 patients treated in London, two were treated conservatively, one of whom stopped using cocaine and symptoms settled. Three patients were treated with oral prednisolone. All three continued to have symptoms, although there was some partial improvement. Two of these three patients reported persistent use of cocaine. Seven patients received prednisolone and rituximab, of whom four improved, including the patient with renal disease. All four patients who improved reported stopping cocaine usage. Three of the seven patients who were treated with prednisolone and rituximab reported no improvement, but all three continued to use cocaine. Finally, one patient received MTX for RA but was lost to follow-up.

Discussion

This study represents the largest cohort of patients with cocaine-induced vasculitis in the UK, and our findings make a significant contribution to the existing literature. Using a systematic approach in our Birmingham clinic, we identified 28 patients over a 5-year period who presented with symptoms compatible with GPA associated with cocaine usage. Nine of these patients denied any cocaine use. Failure to undertake urine toxicology would have resulted in a missed diagnosis in 32% of our patients, who probably would have gone on to receive treatment with immunosuppression. Using a high index of suspicion, our London colleagues diagnosed 13 patients over the same length of time. Both these approaches reveal considerably higher rates of diagnosis than previously published [17]. Patients with cocaine-induced mimics of GPA are considerably younger, with a median age of 41 years; this is 20 years younger than patients identified with idiopathic GPA and predominantly ENT symptoms [18].

ANCA positivity is well recognized in patients with cocaine-induced disease; only five of our patients were negative for ANCA by either IIF or ELISA. However, as is well described, ANCA positivity is not diagnostic of ANCAassociated vasculitis. Indeed only 2 of 16 patients who had a nasal or sinus biopsy had evidence of vasculitis, and none had granulomata characteristic of GPA.

Our patients were characteristic of patients with cocaineinduced midline destructive lesions, with 30 patients having septal perforations, of whom six had oronasal fistulas. Patients with midline destructive lesions are reported to be PR3-ANCA positive, with a typical pANCA pattern [9]. However, we found that only 29% of our patients had a pANCA pattern with PR3-ANCA positive by ELISA. In a study of 30 patients from the USA, where 70% of cocaine is adulterated with levamisole, MPO-ANCA were reported in all cases, and 50% were dual positive for both MPO and PR3; they considered dual positivity to be pathognomonic of cocaine-induced disease [19]. An additional prospective cohort study comparing 31 patients with cocaine/levamisoleinduced vasculitis and 45 patients with idiopathic vasculitis found that the cohort using cocaine were more likely to be MPO positive (100% vs 67%) [20]. However, none of our patients had MPO-ANCA identified by ELISA. Interestingly, 6 of 149 (4%) patients who were IIF positive from the CYCLOPS trial tested positive for both PR3- and MPO-ANCA [21]. All these patients had renal involvement, which is uncommon in cocaine-induced disease, adding weight to our results that dual positive ANCA is not pathognomonic of cocaine-induced pseudo-vasculitis.

Although our findings are similar to those of a previous publication from London [17], results might differ from other publications because the amount of levamisole present in samples from our patients is unknown. A European analysis suggested that 40-70% of cocaine is contaminated with levamisole but at variable concentrations, with the average cocaine sample analysed containing 50% cocaine and 8% levamisole [22]. This might impact on the autoantibodies detected. Cocaine-induced vasculitis can present with constitutional and systemic symptoms. However, only 28% of our patients had symptoms beyond the upper respiratory tract. Only one of our patients had acute kidney injury, but there was no pulmonary haemorrhage. Our results might differ from other published studies because the majority of our patients were diagnosed through a specialized ENT/vasculitis service and presented with cocaine-induced midline destructive lesion-predominant disease, reflecting the referral pathways, whereas other publications are predominantly from a rheumatological or nephrological pathway [17, 19]. Palatal perforation is rare in idiopathic GPA [11], but 15% of our patients had oronasal fistula. Our data support the view that patients with destructive nasal lesions without significant systemic symptoms should be investigated for cocaine-induced disease, especially if younger than expected.

Treatment approaches

There is significant heterogeneity in the literature regarding treatment of patients with cocaine-induced vasculitis. Corticosteroids (CSs) and immunosuppression with MTX and CYC are often used [17]. The practice in our Birmingham centre is to offer immunosuppressive therapy if the patient has stopped using cocaine and remains symptomatic. Importantly, patients who continued to be followed up and managed to discontinue cocaine use improved without the need for immunosuppression. Our London colleagues have used immunosuppression in the majority of patients. Despite this, patients failed to eradicate symptoms without stopping cocaine use. The experience in our two different centres suggests that discontinuation of cocaine is required to manage patients and that symptoms will persist despite immunosuppression if there is ongoing cocaine use. This is in agreement with case reports in the literature [3, 5, 12, 23–25, 26]. Numerous studies have observed that cessation of cocaine use leads to complete resolution of the condition [3, 23–25]. Furthermore, where immunosuppressive therapies, such as CSs or CYC, were used (mainly in cases of renal involvement), full resolution was not achieved unless the patient stopped using cocaine [25, 27]. In patients who did not stop using cocaine, resolution was never achieved [13, 14]. Immunosuppression is associated with significant risks, particularly infection, and inappropriate use can cause significant harm. Given our evidence of lack of benefit, immunosuppression in the absence of failure to discontinue cocaine should be considered very carefully, and we would suggest not used. Individuals should be referred to addiction counselling services to help with cocaine cessation.

There is evidence to suggest that autoantibody levels normalize within 2–14 months of discontinuing cocaine/levamisole without immunosuppressive treatment [23]. This might reflect laboratory research showing that levamisole activates neutrophils via M_3 muscarinic acetylcholine receptors to extrude their intracellular proteins in neutrophil extracellular traps (NETs) via a process called NETosis, triggering autoantibody formation [10]. Withdrawal of levamisole led to cessation of NETosis and a reduction in autoantibody production without the need for other interventions [10].

Strengths and limitations

This report describes the largest cohort of patients in the UK with cocaine-induced vasculitis and emphasizes the importance of performing toxicology on patients who present with isolated nasal symptoms that might be consistent with vasculitis. The different treatment approaches between the two centres allows us to compare outcomes between the two strategies, emphasizing the need for cocaine discontinuation and conservative management.

There were limitations to the study, including its retrospective nature and a high proportion of patients lost to follow-up. We used urine toxicology testing routinely in only one centre, which might therefore underestimate the number of patients with this diagnosis. Given that the majority of patients came via an ENT referral pathway, this might skew the number of patients with the cocaine-induced midline destructive lesions pattern of disease, with under-representation of those with more systemic disease. We were unable to report the longitudinal evaluation of how ANCA levels or binding patterns changed owing to the high proportion of patients lost to follow-up.

Conclusion

Our data show that cocaine-induced vasculitis is more common than first reported and that toxicology should be considered on all patients who appear to have isolated nasal involvement with vasculitis. We argue that MPO-ANCA positivity might not be as common as previously reported and that dual positive ANCA is uncommon, and therefore, the lack of dual positive PR3- and MPO-ANCA should not be used to exclude cocaine-induced disease. We are reassured to report that renal and pulmonary involvement appears to be rare. Furthermore, we advocate for cessation of cocaine use as the first strategy, without use of immunosuppression, for the management of these patients.

Data availability

The data underlying this article are available in the article.

Funding

No specific funding was received from any bodies in the public, commercial or not-for-profit sectors to carry out the work described in this article.

Disclosure statement: The authors have declared no conflicts of interest.

References

- United Kingdom drug situation 2019: summary. London: GOV.UK, 2021. https://www.gov.uk/government/publications/ united-kingdom-drug-situation-focal-point-annual-report/uk-drugsituation-2019-summary#:~:text=The%20most%20commonly% 20used%20drugs,%2C%20MDMA%20%2C%20ketamine% 20and%20amphetamine (9 June 2022, date last accessed).
- Trimarchi M, Gregorini G, Facchetti F *et al.* Cocaine-induced midline destructive lesions. Medicine (Baltimore) 2001;80:391–404.
- Jin Q, Kant S, Alhariri J, Geetha D. Levamisole adulterated cocaine associated ANCA vasculitis: review of literature and update on pathogenesis. J Community Hosp Intern Med Perspect 2018;8:339–44.
- 4. Molloy ES, Langford CA. Vasculitis mimics. Curr Opin Rheumatol 2008;20:29–34.

- 5. Nolan AL, Jen K-Y. Pathologic manifestations of levamisoleadulterated cocaine exposure. Diagn Pathol 2015;10:48.
- Armstrong M, Shikani AH. Nasal septal necrosis mimicking Wegener's granulomatosis in a cocaine abuser. Ear Nose Throat J 1996;75:623–6.
- Lötscher F, Krusche M, Ruffer N *et al.* Cocaine-induced ANCAassociated renal disease: a case-based review. Rheumatol Int 2019; 39:2005–14.
- Lood C, Hughes GC. Neutrophil extracellular traps as a potential source of autoantigen in cocaine-associated autoimmunity. Rheumatology (Oxford) 2017;56:638–43.
- Wiesner O, Russell KA, Lee AS *et al.* Antineutrophil cytoplasmic antibodies reacting with human neutrophil elastase as a diagnostic marker for cocaine-induced midline destructive lesions but not autoimmune vasculitis. Arthritis Rheum 2004;50:2954–65.
- Carmona-Rivera C, Purmalek MM, Moore E et al. A role for muscarinic receptors in neutrophil extracellular trap formation and levamisole-induced autoimmunity. JCI Insight 2017;2:e89780.
- Trimarchi M, Bondi S, Dela Torre E, Terreni MR, Bussi M. Palate perforation differentiates cocaine-induced midline destructive lesions from granulomatosis with polyangiitis. Acta Otorhinolaryngol Ital 2017;37:281–5.
- 12. Souied O, Baydoun H, Ghandour Z, Mobarakai N. Levamisolecontaminated cocaine: an emergent cause of vasculitis and skin necrosis. Case Rep Med 2014;2014:434717.
- Baptiste GG, Alexopoulos A-S, Masud T, Bonsall JM. Systemic levamisole-induced vasculitis in a cocaine user without cutaneous findings: a consideration in diagnosis. Case Rep Med 2015;2015: e547023.
- Kumar D, Batal I, Jim B, Mendez B, Anis K. Unusual case of levamisole-induced dual-positive ANCA vasculitis and crescentic glomerulonephritis. BMJ Case Rep 2018. Published on 20 August 2018, doi: 10.1136/bcr-2018-225913.
- Moinuddin I, Madhrira M, Bracamonte E, Thajudeen B, Sussman A. Membranous nephropathy with crescents associated with levamisole-induced MPO-ANCA vasculitis. Pathol Res Pract 2016; 212:650–3.

- Pendergraft WF, Niles JL. Trojan horses: drug culprits associated with antineutrophil cytoplasmic autoantibody (ANCA) vasculitis. Curr Opin Rheumatol 2014;26:42–9.
- 17. Subesinghe S, van Leuven S, Yalakki L, Sangle S, D'Cruz D. Cocaine and ANCA associated vasculitis-like syndromes a case series. Autoimmun Rev 2018;17:73–7.
- 18. Thayakaran R, Goel R, Adderley NJ *et al.* Cluster analysis of patients with granulomatosis with polyangiitis (GPA) based on clinical presentation symptoms: a UK population-based cohort study. Arthritis Res Ther 2022;24:201.
- McGrath M, Isakova T, Rennke HG *et al.* Contaminated cocaine and antineutrophil cytoplasmic antibody-associated disease. Clin J Am Soc Nephrol 2011;6:2799–805.
- Emil NS, Cisneros DR, Penmetsa S *et al.* Atypical chronic inflammatory MNCA-positive deforming arthritis after cocainelevamisole exposure. J Clin Rheumatol 2020;26:24–32.
- de Groot K, Harper L, Jayne DRW *et al.* Pulse versus daily oral cyclophosphamide for induction of remission in antineutrophil cytoplasmic antibody-associated vasculitis: a randomized trial. Ann Intern Med 2009;150:670–80.
- 22. Brunt TM, Nagy C, Bücheli A *et al.* Drug testing in Europe: monitoring results of the Trans European Drug Information (TEDI) project. Drug Test Anal 2017;9:188–98.
- Gross RL, Brucker J, Bahce-Altuntas A *et al*. A novel cutaneous vasculitis syndrome induced by levamisole-contaminated cocaine. Clin Rheumatol 2011;30:1385–92.
- 24. Gill H, Trinh D, Anderson DJ, Li N, Madenberg D. Cocaine and levamisole induced vasculitis. Cureus 2021;13:e17192.
- Latif Z, Pour-Ghaz I, Bergeron JB. Levamisole-induced vasculitis in a hepatitis C patient: a general medicine ward perspective on diagnosis and management. Cureus 2019;11:e5198.
- Pham T, Heinly C, Herman C, Selim MA. P-ANCA positive cocaine associated vasculitis: a case report. J Cutan Pathol 2005;32:109.
- Olives TD, Kornas RL, Fujisawa R, Cole JB. Unexpected complication of cocaine-associated anti-neutrophil cytoplasmic antibody vasculitis related to persistent in-hospital cocaine use. J Addict Med 2017;11:157–60.



A 2nd generation, JAK1 preferential inhibitor for moderate to severe RA¹⁻⁶

While 1st generation JAK inhibitors are relatively non-selective,²⁻⁶ JYSELECA has over 5x greater potency for JAK1 over JAK2/3 and TYK21*

Balancing sustained efficacy⁷⁻¹¹ with acceptable tolerability^{1,12}



()

Indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti-rheumatic drugs.¹ May be used as monotherapy or in combination with methotrexate.¹

*From biochemical assays, the clinical relevance of which is uncertain. JAK, Janus kinase; RA, rheumatoid arthritis; TYK, tyrosine kinase.

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information.

۲

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information. **JYSELECN** figotinib 100 mg or 200 mg film-coated tablets. **Indication:** Jyseleca is indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti rheumatic drugs (DMARDs). Jyseleca may be used as monotherapy or in combination with methotrexate (MTX). **Dosage:** <u>Adults</u>; 200 mg once daily. Taken orally with/without food. It is recommended that tablets are swallowed whole. <u>Laboratory Monitoring:</u> Refer to the SmPC for information regarding laboratory monitoring and dose initiation or interruption. <u>Elderly</u>, 4 starting dose of 100 mg of filgotinib once daily is recommended for patients aged 75 years and older as clinical experience is limited. <u>Renal impairment</u>: No dose adjustment required in patients with estimated creatinine clearance (CrCl) ≥ 60 mL/min. A dose of 100 mg of filgotinib once daily is recommended for patients with estimated and by is recommended for patients. Safety and efficacy not yet established. **Contraindications**: Hypersensitivity to the active substance or to any of the excipients. Active tuberculosis (TB) or active serious infections. Pregnancy. **Warnings/Precautions**: See SmPC for full information. <u>Immunosuppression</u>: combination use, with immunosuppressints (AK) inhibitors is not recommended as a risk of additive immunosuppressions infections such as pneumonia and opportunistic infections equipations: thypersensitivity to the active sub excluded. <u>Infections</u>; Infections, including serious infections, Pregnancy. **Warnings/Precautions**: Shave been reported. Risk beneft should be assessed prior to initiating in patients with risk factors for infections (see SmPC). Patients should be closely monitored for the development of the initiation in the advelopment of the development of the development of the material and opportunistic infections equipations have been reported, Kisk benefit should be assessed phore of hitating in patients with risk factors for infections (see SmPC). Yatients should be closely monitored for the development of igns and symptoms of infections during and after fligotinib reatment. Treatment should be interrupted if the patient

is not responding to antimicrobial therapy, until infection is controlled. There is a higher incidence of serious infections in the elderly aged 75 years and older, caution should be used when treating this population. <u>Tuberculosis</u>, Patients should be screened for TB before initiating filgotinib, and filgotinib should not be administered to patients with active TE. <u>Viral</u> <u>reactivation</u>: Cases of herpes virus reactivation (e.g., herpes zoster), were reported in clinical studies (see SmPC). If a patient develops herpes zoster, filgotinib treatment should be temporarily interrupted until the episode resolves. Screening for viral hepatitis and monitoring for reactivation should be performed. <u>Malignancy</u>: Immunomodulatory medicinal products may increase the risk of malignancies. Malignancies were observed in clinical studies (see SmPC). <u>Ferlility</u>. In animal studies, decreased ferlility, impaired spermatogenesis, and histopathological effects on male reproductive organs were observed (see SmPC). The potential effect of filgotinib on sperm production and male fertility in humans is currently unknown. <u>Haematological abnormalities</u>; Do not start therapy, or temporarily stop, if Absolute Neutrophil Count (ANC) 1< 10° (cells/L, ALC - OS + 10° cells/L or chaemoglobin «B g/dL. Temporarily stop therapy if these values are observed during routine patient management. <u>Vaccinations</u>; Use of live vaccines during, or immediately prior to, filgotinib treatment is not recommended. <u>Lipids</u>: Treatment with filgotinib was associated with dose dependent increases in lipid parameters, including total cholesterol, and high-density lipoprotein (LDL) levels, while low density lipoprotein (LDL) levels were slightly increased (see SmPC). <u>Cardiovascular</u> *tisk*; Rheumatoid arthritis patients have an increased insk for cardiovascular disorders. Patients should have risk factors (e.g., hypertension, hyperlipidaemia) managed as part of usual standard of care. <u>Venous thrombobembolism</u>: Events of deep venous thrombosis (OVT) and pulmona of DVT/PE, or patients undergoing surgery, and prolonged

Learn more at strengthofbalance.co.uk

immobilisation. Lactose content: Contains lactose; patients with rare hereditary problems of galactose intolerance, total lactase deficiency or glucose-galactose malabsorption should not take filgotinib. **Pregnancy/Lactation**: Filgotinib is contraindicated in pregnancy. Filgotinib should not be used during breast-feeding. Women of childbearing potential must use effective contraception during and for at least 1 week after cessation of treatment. **Driving/Using machinery**: No or negligible influence, however dizziness has been reported. **Side effects:** See SmPC for full information. <u>Common (21/100)</u>: herpes zoster, pneumonia, neutropenia, hypercholesterolasemia infection and dizziness. <u>Uncommon (s1/1000 to 1/100)</u>; herpes zoster, pneumonia, neutropenia, hypercholesterolaemia and blood creatine phosphokinase increase. Serious side effects: See SmPC for full information **Legal category**: POM **Pack**: 30 film-coated tablets/bottle **Price**: UK Basic NHS cost: £863.10 Marketing authorisation number(5): <u>Great Britain</u> Jyseleca 100mg film-coated tablets PLGB 42147/0002 hypeleca 200mg film-coated tablets PLGB 42147/0002 hypeleca 200mg film-coated tablets PLGB 42147/0002 hypeleca 100mg film-coated tablets EU/1/20/1480/001 EU/1/20/1480/003 EU/1/20/1480/004 **Further information**: Galapagos UK, Belmont House, 148 Belmont Road, Uxbridge (DB8 105, United Kingdom 00800 7387 1345 **medicalinfo@glgg**. <u>com</u> Jyseleca[®] is a trademark. **Date of Preparation**: January 2022 UK-RA-FIL-202201-00019 **W** Additional monitoring required Additional monitoring required

Adverse events should be reported. For Great Britain and Northern Ireland, reporting form and information can be found at <u>yellowcard.mhra.gov.</u> and information can be found at <u>yellowcard.mnra.gov.u</u> or via the Yellow Card app (download from the Apple Ap Store or Google Play Store). Adverse events should also be reported to Galapagos via email to DrugSafety.UK.Ireland@glpg.com or 00800 7878 1345

References: 1. JYSELECA SPC. Available at: www.medicines.org.uk. Last accessed: June 2022. 2. Angelini J, et al. Biomolecules 2020;10(7):E1002. 3. Banerjee S, et al. Drugs 2017;77:521–546. 4. O'Shea JJ, et al. Nat Rev Rheumatol 2013;9(3):173–182. 5. Traves PG, et al. Ann Rheum Dis 2021;0(1-11. 6. McInnes IB, et al. Arthr Res Ther 2019;21:183. 7. Combe B, et al. Ann Rheum Dis 2021;doi:10.1136/ annrheumdis-2020-219214. 8. Genovese MC, et al. JAMA 2019;322 (4):315–325. 9. Westhowers R, et al. Ann Rheum Dis 2021;doi:10.1136/annrheumdis-2020-219213. 10. Combe B, et al. Arthritis Rheumatol 2021;73(suppl 10). https://acrabstract/clinical-ou-week-48-of-fig0tinib-treatment-in-an-ongoing-long-term-extension-trial-of-a-plates-twith-inadequate-response-to-mtx-initially-treated-with-filgotinib-ra-adalimumab-during-th/. Last accessed: June 2022. 11. Buch MH, et al. Arthritis Rheumatol 2021;73 (suppl 10). https://acrabstracts.org/abstract/clinical-outcomes-up-to-week-48-of-ongoing-filgotinib-tra-abstracts.org/abstract/Jactastracts.org/abstract/sorg/abstract/linical-outcomes-up-to-week-48-of-Ingoing-filgotinib-tra-abstracts.org/abstract/linical-outcomes-up-to-week-48-of-ongoing-filgotinib-tra-abstracts.org/abstract/linical-outcomes-up-to-week-48-of-ongoing-filgotinib-tra-abstracts.org/abstract/lineg/abstract/linegrated-safety-analysis-update-for-filgotinib-in-patients-with-moderately-to-severely-active-rheumatoid-arthritis-receiving-treatment-over-a-median-of-2-2-years/. Last accessed: June 2022.



June 2022 GB-RA-JY-202205-00033

JYSELECA, GALAPAGOS and the JYSELECA and GALAPAGOS logos are registered trademarks of Galapagos NV. © 2022 Galapagos NV. All rights reserved.

۲